
 
PERFORMANCE ENHANCEMENT AND EMISSIONS  

REDUCTION IN JAGGERY PLANTS THROUGH CLEAN  
COMBUSTION USING PROCESSED AGRO-RESIDUE  

WASTES 
 
 
 
 
 

ALOK KUMAR 
 
 
 
 

 
 
 

 

 

DEPARTMENT OF ENERGY SCIENCE AND ENGINEERING 
INDIAN INSTITUTE OF TECHNOLOGY DELHI 

AUGUST 2025 



 

 

 

 

 

 

 

 

 

 

 

 

 

 

 

© Indian Institute of Technology Delhi (IITD), New Delhi, 2025 

 



Performance Enhancement and Emissions Reduction in 

Jaggery Plants Through Clean Combustion using 

Processed Agro-residue Wastes 

by 

ALOK KUMAR 

Department of Energy Science and Engineering 

Submitted 

in fulfilment of the requirements of the degree of Doctor of Philosophy 

to the 

INDIAN INSTITUTE OF TECHNOLOGY DELHI 

August 2025



Dedicated 

to 

My Beloved Parents 



DECLARATION 

 I hereby declare that the work presented here in the thesis entitled “Performance 

Enhancement and Emissions Reduction in Jaggery Plants Through Clean Combustion Using 

Processed Agro-residue Wastes” has been carried out by me towards the partial fulfilment of 

the requirement for the award of the degree of Doctor of Philosophy at the Department of 

Energy Science and Engineering, Indian Institute of Technology Delhi. The content of this 

thesis, in full or in parts, has not been submitted to any other Institute or University for the 

award of any degree or diploma by me. 

Alok Kumar 

Entry No.: 2020ESZ8156

E-mail:esz208156@iitd.ac.in

Place: New Delhi  

i



CERTIFICATE 

 This is to certify that the thesis entitled “Performance Enhancement and Emissions 

Reduction in Jaggery Plants Through Clean Combustion Using Processed Agro-residue 

Wastes” being submitted by Mr. Alok Kumar in fulfilment of the requirements for the award 

of the degree of Doctor of Philosophy is a record of bonafide research work performed by 

him under our guidance and supervision at Department of Energy Science and

Engineering, Indian Institute of Technology Delhi.  

 Further, the results obtained herein have not been submitted in part or in full to any other 

University or Institute for the award of any degree to the best of our knowledge.

Prof. S. K. Tyagi 

Professor 

Department of Energy Science and Engineering 

Indian Institute of Technology Delhi

Prof.  K. Agrawal 

Assistant Professor 

Department of Mechanical Engineering 

Indian Institute of Technology Delhi

ii



ACKNOWLEDGEMENTS 

 It gives me immense pleasure to express my gratitude to all those who supported me during 

the course of my study. First and foremost, I would like to express my deep gratitude to my 

research supervisor (s), Prof. S. K. Tyagi and Prof. K. Agrawal, for providing me with 

thorough guidance, encouragement, and invaluable suggestions throughout my research work 

despite their busy academic and professional schedules.  

 I would like to extend my sincere thanks to the Student Research Committee (SRC) 

Chairman Prof. Dibakar Rakshit, external member Prof. Upasna Sharma, and internal member 

Prof. Rahul Goyal for providing their valuable suggestions during the various evaluation 

stages of my research work. 

 I acknowledge CRF, IIT Delhi for BET, and DSC. Special thanks to IRD IIT Delhi for 

providing the financial support and instruments facility through the FIRP scheme. Also, 

thanks to PGS&R IIT Delhi for the RSTA to attend an international conference in Gyor, 

Hungary. I also acknowledge the jaggery unit owner (Shri Narendra Sharma) for providing 

the opportunity to carry out the experimental investigation at the four-pan jaggery unit 

which is located at village Basaud in district Baghpat of Uttar Pradesh, India. 

 Special thanks to Dr. Devershi Morya, Dr. Himanshu, Dr. Abhishek Verma, Dr. 

Sandeep Singh, Dr. Prathvi Raj Chauhan, Mr. Raveesh G, Mr. Sanjaya Sena, Mr. Siva P, Ms. 

Riya Sharma, and Mr. Sachin Kamboj for their unwavering support during my study at 

Solar and Biomass Thermal Science Research Laboratory at DESE, IIT Delhi. 

 I sincerely express my heartfelt gratitude to my parents, my elder brother, Shri Santosh 

Kumar (IPS), my younger brother, Dr. Nishant Raj, and my beloved wife, Dr. Akanksha 

Neha, for their unwavering support, constant encouragement, and blessings, which have been 

instrumental in the successful completion of my research work.  

(Alok Kumar) 

iii



ABSTRACT 

Jaggery, commonly known as 'Gur' in the Indian subcontinent, is a traditional 

sweetener produced by concentrating the juice extracted from sugarcane or palm trees into a 

solid or semi-solid form. The fibrous residue obtained from sugarcane crushing, known as 

bagasse, is sun-dried and subsequently used as a primary fuel source in the jaggery production 

process. The traditional jaggery production, while culturally significant and accessible, suffers 

from low thermal efficiency, high emissions, and excessive bagasse consumption. These 

limitations highlight the need for technological interventions aimed at improving energy 

efficiency, reducing environmental impact, and optimizing resource utilization in the jaggery 

production process. This thesis delves into a comprehensive experimental and numerical 

investigation into the thermal performance and gaseous emissions of traditional and modified 

four-pan jaggery production units.  The present study aims to assess the energy and exergy 

efficiencies based on resource consumption and transformations occurring in the sugarcane-

based jaggery manufacturing process. Initially, an exergy analysis of the traditional jaggery 

unit is conducted using experimental data, thermodynamic relations, and specific exergy values 

for the different system components. This analysis is performed for both summer and winter 

seasons. The results revealed that the exergy sustainability scores are low during both seasons, 

indicating significant environmental impacts and inefficiencies in the process. The 

experimental energy and exergy efficiencies are found to be higher in the summer season 

compared to winter. Specifically, the energy efficiency is recorded at 25.72% in summer and 

17.62% in winter, while the exergy efficiency of bagasse combustion is 3.69% in summer and 

2.89% in winter. These differences are attributed to higher ambient temperatures, lower 

moisture content in bagasse, and greater brix content in sugarcane during the summer. The 

exergy destruction due to combustion is significant, calculated as 85.28% in winter and 85.77% 
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in summer. This is primarily due to thermal irreversibilities in combustion and heat transfer 

processes. The operation of the four-pan jaggery unit is found to be more eco-friendly during 

the summer due to a higher Exergy-Based Index (EBI). However, the sustainability index 

remained low for both seasons, suggesting that although renewable energy sources are used, 

the environmental emissions are considerable. To improve exergy efficiency, sustainability 

index, and EBI, it is necessary to reduce the destruction number (DN), environmental 

destruction coefficient (EDC), and exergy destruction index (EDI). The emissions data for the 

traditional jaggery unit showed that carbon monoxide (CO) levels ranged from 23,500 to 

65,000 ppm, nitrogen oxides (NOx) from 70 to 855 ppm, and sulphur dioxide (SO2) from 1,720 

to 3,350 ppm.  

A computational fluid dynamics (CFD) model is developed and validated against 

experimental results for traditional unit, demonstrating good agreement in terms of thermal 

efficiency and flue gas temperature profiles. Several design variations are analysed 

numerically, including changes to the inlet position and area, the application of thermal 

insulation, the implementation of a deep-pan strategy, and zig-zag furnace flow design. The 

study found that radiative heat transfer is the dominant mode of heat transfer to the pans.  

Alterations to the inlet area and location had minimal impact on heat transfer to the pans due 

to the minor role of convection in this process. This finding differs from previous literature, 

likely due to the higher flue gas temperatures in the present setup. The flow dynamics are not 

significantly affected by these geometrical changes. The deep-pan strategy showed significant 

potential for enhancing thermal efficiency, which increased to 26%, marking a 30–40% 

improvement over the baseline unit. This improvement is mainly due to a larger radiative heat 

exchange surface. Additionally, Pan 4, which is the first to receive the flue gases, experienced 

increased heat transfer rate due to flow impingement and enhanced turbulence caused by cavity 

formations between the pans. Thermal insulation using ceramic blankets is also studied, with 
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thicknesses ranging from 1 to 4 inches. The analysis found that a 2-inch thickness is optimal, 

providing effective thermal retention with no significant gains beyond this point. Insulation at 

the top furnace wall near the pan sides is found to be most effective. When combined with the 

deep-pan strategy, thermal insulation further improved performance. Specifically, the thermal 

efficiency increased to 25.7% with 2-inch insulation and to 26.1% with 4-inch insulation, 

compared to the baseline of 18.82%. However, due to cost considerations, 2-inch insulation is 

deemed optimal. A zig-zag flow furnace design is also tested, with multiple slot hole 

geometries. These changes showed minimal impact on performance in the absence of 

insulation. However, when combined with insulation and the deep-pan strategy, the zig-zag 

configuration achieved a thermal efficiency of 25.12% without insulation and 26.60% with 

insulation. In this design strategy, convective heat transfer became the dominant mode, which 

is a contrast to the baseline case where radiative heat transfer dominated. 

An experimental evaluation of a burner-integrated modified jaggery unit, insulated with 

a 2-inch ceramic blanket, showed substantial improvement in thermal performance. The 

thermal efficiency ranged from 42% to 45%, while emissions are significantly reduced. The 

CO emissions dropped to 120–300 ppm, NOx to 171–785 ppm, and SO2 to 1.50–121 ppm. 

Numerical simulations for this modified unit showed strong agreement with experimental data 

in terms of both thermal efficiency and slot hole temperature profiles. Like the zig-zag furnace, 

the modified unit exhibited a dominance of convective heat transfer. Further numerical studies 

on the modified jaggery unit incorporating deep-pan and mixed-pan strategies revealed 

remarkable thermal performance improvements. The deep-pan strategy increased thermal 

efficiency to 53.1%, while the mixed-pan strategy achieved a peak efficiency of 63.27%, 

underscoring the effectiveness of combining structural optimization with thermal management. 
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सार 

गुड़, जिसे भारतीय उपमहाद्वीप में सामान्यतः  'गुड़' कहा िाता है, एक पारंपररक जमठाई है जिसे 

गने्न या ताड़ के पेड़  ंसे जिकाले गए रस क  गाढा करके ठ स या अर्ध-ठ स रूप में तैयार जकया िाता 

है। गने्न की पेराई से प्राप्त रेशेदार अवशेष, जिसे बगास कहा िाता है, क  रू्प में सुखाकर गुड़ उत्पादि 

प्रजिया में प्रमुख ईंर्ि के रूप में उपय ग जकया िाता है। हालााँजक पारंपररक गुड़ उत्पादि सांसृ्कजतक 

रूप से महत्वपूर्ध और सुलभ है, लेजकि इसमें तापीय दक्षता कम ह ती है, उत्सिधि अजर्क ह ता है और 

बगास की अत्यजर्क खपत ह ती है। इि सीमाओ ंक  देखते हुए, ऊिाध दक्षता बढािे, पयाधवरर्ीय प्रभाव  ं

क  कम करिे और संसार्ि  ंके इष्टतम उपय ग के जलए तकिीकी हस्तके्षप की आवश्यकता है। यह 

श र् कायध पारंपररक और पररवजतधत चार-कढाही वाले गुड़ इकाइय  ं की तापीय प्रदशधि और गैसीय 

उत्सिधि  ंका एक व्यापक प्रय गात्मक एवं संख्यात्मक (numerical) अध्ययि प्रसु्तत करता है। अध्ययि 

का उदे्दश्य गने्न पर आर्ाररत गुड़ जिमाधर् प्रजिया में ऊिाध और एक्सिी दक्षताओ ंका संसार्ि खपत 

और ह िे वाले रूपांतरर्  ं के आर्ार पर मूल्ांकि करिा है। प्रारंभ में पारंपररक गुड़ इकाई की 

एक्सिी जवशे्लषर् जकया गया, जिसमें प्रय गात्मक आंकड़ ,ं उष्मागजतकी (Thermodynamic) 

समीकरर्  ं और जवजभन्न प्रर्ाली घटक  ं के जलए जवजशष्ट एक्सिी माि  ं का उपय ग जकया गया। यह 

जवशे्लषर् ग्रीष्मकाल और शीतकाल द ि  ंऋतुओ ंके जलए जकया गया। पररर्ाम  ंसे पता चला जक द ि  ं

मौसम  ंमें एक्सिी स्थाजयत्व स्क र (Exergy Sustainability Score) कम था, जिससे पता चलता है जक 

प्रजिया में पयाधवरर्ीय प्रभाव और अक्षमताएाँ  काफी अजर्क हैं। प्रय गात्मक रूप से ऊिाध और एक्सिी 

दक्षताएाँ  गजमधय  ंमें सजदधय  ंकी तुलिा में अजर्क पाई गईं। गजमधय  ंमें ऊिाध दक्षता 25.72% और सजदधय  ं

में 17.62% दिध की गई, िबजक बगास दहि की एक्सिी दक्षता गजमधय  ं में 3.69% और सजदधय  ं में 

2.89% रही। यह अंतर उच्च पररवेश तापमाि, बगास में कम िमी और गने्न में अजर्क जिक्स कंटेंट के

कारर् था। दहि के कारर् एक्सिी िष्ट ह िा काफी अजर्क था — सजदधय  ंमें 85.28% और गजमधय  ंमें 
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क् जंक इसमें उच्च एक्सिी आर्ाररत सूचकांक (EBI) दिध हुआ। हालााँजक, स्थाजयत्व सूचकांक 

(Sustainability Index) द ि  ं मौसम  ं में कम रहा, यह दशाधता है जक िवीकरर्ीय ऊिाध स्र त  ं के

उपय ग के बाविूद पयाधवरर्ीय उत्सिधि पयाधप्त है। एक्सिी दक्षता, स्थाजयत्व सूचकांक और EBI में 

सुर्ार के जलए Destruction Number (DN), Environmental Destruction Coefficient (EDC) और 

Exergy Destruction Index (EDI) क  कम करिा आवश्यक है। पारंपररक गुड़ इकाई के जलए 

उत्सिधि डेटा के अिुसार काबधि म ि ऑक्साइड (CO) 23,500 से 65,000 ppm, िाइटर  िि ऑक्साइड 

(NOx) 70 से 855 ppm, और सल्फर डाइऑक्साइड (SO2) 1,720 से 3,350 ppm तक दिध जकया गया। 

एक पारंपररक इकाई के जलए जवकजसत जकया गया कम्प्यूटेशिल फू्लड डायिेजमक्स (CFD) 

मॉडल प्रय गात्मक पररर्ाम  ं से अच्छी तरह मेल खाता है, जवशेषकर तापीय दक्षता और फ्ल्लू् गैस 

तापमाि प्र फाइल के संदभध में। कई जडजाइि बदलाव  ंका संख्यात्मक जवशे्लषर् जकया गया — िैसे 

जक इिलेट की स्थस्थजत और के्षत्र में बदलाव, थमधल इंसुलेशि का उपय ग, डीप-पैि रर्िीजत और जजग-

जैग भट्ठी प्रवाह जडजाइि। अध्ययि से यह पाया गया जक पैि  ं क  ऊष्मा स्थािांतरर् में जवजकरर्ीय 

ऊष्मा (Radiative Heat Transfer) प्रमुख भूजमका जिभाती है। इिलेट के्षत्र और स्थस्थजत में बदलाव से 

ऊष्मा स्थािांतरर् पर जवशेष प्रभाव िही ंपड़ा क् जंक संवहि (Convection) का य गदाि नू्यितम पाया 

गया। यह जिष्कषध जपछले साजहत्य से अलग है, संभवतः  वतधमाि प्रर्ाली में उच्च फ्ल्लू् गैस तापमाि के

कारर्। डीप-पैि रर्िीजत िे तापीय दक्षता में उले्लखिीय सुर्ार जदखाया, ि  बढकर 26% ह  गई — 

ि  जक बेसलाइि इकाई की तुलिा में 30–40% अजर्क है। इसका मुख्य कारर् बढी हुई जवजकरर्ीय 

ऊष्मा जवजिमय सतह है। इसके अजतररक्त, पैि 4 — ि  सबसे पहले फ्ल्लू् गैस प्राप्त करता है — में 

कैजवटी जिमाधर् और प्रवाह टकराव के कारर् ऊष्मा स्थािांतरर् दर अजर्क रही। जसरेजमक ब्लैंकेट से 

थमधल इंसुलेशि (1 से 4 इंच म टाई) का भी जवशे्लषर् जकया गया। 2 इंच की म टाई क  सबसे उपयुक्त 

85.77% — ि  मुख्यतः  दहि और ऊष्मा स्थािांतरर् प्रजियाओ ं में तापीय अपाररवतधिीयताओ ं के

कारर् था। चार-कढाही प्रर्ाली का संचालि ग्रीष्मकाल में अजर्क पयाधवरर्-अिुकूल पाया गया, 
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4-इंच के साथ 26.1% ह  गई, िबजक बेसलाइि मात्र 18.82% थी। हालांजक लागत क  देखते हुए 2-इंच

इंसुलेशि क  उपयुक्त मािा गया। जजग-जैग फ्ल  जडजाइि का भी परीक्षर् जकया गया, जिसमें कई 

स्लॉट ह ल ज्य मेटर ी शाजमल थी।ं केवल जजग-जैग बदलाव से खास लाभ िही ं जमला, लेजकि िब इसे 

इंसुलेशि और डीप-पैि रर्िीजत के साथ ि ड़ा गया, त  जबिा इंसुलेशि के दक्षता 25.12% और 

इंसुलेशि के साथ 26.60% तक पहुाँच गई। इस जडिाइि में संवहि ऊष्मा स्थािांतरर् प्रमुख बि गया, 

ि  जक बेसलाइि स्थस्थजत के जवपरीत है िहााँ जवजकरर्ीय स्थािांतरर् प्रमुख था। 

एक बिधर-युक्त संश जर्त गुड़ इकाई (2-इंच जसरेजमक इंसुलेशि सजहत) के प्रय गात्मक मूल्ांकि 

में तापीय प्रदशधि में उले्लखिीय सुर्ार देखा गया। इसकी तापीय दक्षता 42% से 45% रही और 

उत्सिधि में भी भारी कमी आई। CO उत्सिधि 120–300 ppm, NOx 171–785 ppm और SO2 1.50–

121 ppm तक जगर गया। संश जर्त इकाई के जलए जकए गए संख्यात्मक अिुकरर् (Simulation) 

प्रय गात्मक आंकड़  ंसे अच्छी तरह मेल खाते हैं। जजग-जैग भट्ठी की तरह ही, इस इकाई में भी संवहि 

ऊष्मा स्थािांतरर् प्रमुख पाया गया। संश जर्त इकाई में डीप-पैि और जमक्स-पैि रर्िीजतय  ं क 

शाजमल कर जकए गए अजतररक्त संख्यात्मक अध्ययि  ंिे और भी बेहतर तापीय प्रदशधि जदखाया। डीप-

पैि रर्िीजत से तापीय दक्षता 53.1% तक पहुाँची, िबजक जमक्स-पैि रर्िीजत िे 63.27% तक की 

दक्षता हाजसल की, ि  यह दशाधता है जक संरचिात्मक अिुकूलि और थमधल प्रबंर्ि का संयुक्त प्रभाव 

अत्यजर्क प्रभावी है। 

पाया गया क् जंक इससे पयाधप्त ऊष्मा संरक्षर् जमला और उससे अजर्क म टाई से क ई महत्वपूर्ध लाभ 

िही ंजमला। पैि के पास ऊपरी भट्ठी दीवार पर इंसुलेशि सबसे प्रभावी रहा। िब इसे डीप-पैि रर्िीजत 

के साथ ि ड़ा गया, त  तापीय दक्षता में और सुर्ार हुआ: 2-इंच इंसुलेशि के साथ दक्षता 25.7% और 
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